\documentstyle[12pt,fullpage,psfig]{article}

\newcommand{\lecturetitle}[4]{
  \noindent
  CS 280, Computer Vision (Prof. Jitendra Malik) \hfill #3\\
  Lecture \##1 (#2) \hfill {\bf DRAFT} Notes by #4\\[-.05in]
  \mbox{}\hrulefill \mbox{}\\}

\begin{document}

% MODIFY THE FOLLOWING TO INCLUDE THE LECTURE #, TOPIC, DATE, and YOUR NAME
\lecturetitle{1}{Introduction}{Jan 20, 1999}{Monica Lewinsky}

% INCLUDE AN OUTLINE OF THE CLASS TOPICS
\begin{itemize} 
  \item Topic A

  \item Topic B
\end{itemize}

% THE NOTES WILL GO HERE

\section{Topic A}

\subsection{sub topic } 

\subsection{sub topic }

\section{Topic B}

% EXAMPLE OF A PostScript FIGURE
%\begin{figure}
%\centerline{\psfig{file=lec4.eps,height=3in}}
%\caption{Illustration}
%\label{surface}
%\end{figure}


......

\end{document}

